VC- /Week 4 starting   22 Jan 07
SM18 Operation Meeting (Tuesdays) 23 Jan  07

Bench status  on Monday 22-Jan-2007 at 1/22/2007 10:23:47 AM 
	4
	A1
	MBBR2384
	PT 14 Magnet MAPS
	296.88

	12
	A2
	MBAR2326
	ICS 1 WP04 HV Test Warm
	296.88

	10
	B1
	SSS361
	PT5.4: Powering Corrector EF
	1.90

	2
	B2
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	297.41

	11
	C1
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	295.8

	5
	C2
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	296.25

	6
	D1
	SSS523
	PT 12 Warm Up
	286.25

	9
	D2
	SSS509
	PT 13.2 Resist. Meas.
	296.88

	1
	E1
	MBAR2414
	PT 10 Min. NRJ 11850 A
	32.63

	7
	E2
	MBBR2423
	ICS 1 WP04 HV Test Warm
	296.43

	3
	F1
	MBAR2328
	ICS 3 Traveller after warm test
	0

	8
	F2
	MBAR2408
	PT 5.2 1.5 KA Quench HF
	0


  
CONFIG :  D 6 - AS 2 -  IRSSS 4 ;   
Supervisor: Marco; then, Sandrine(29/1),  Stephane(5/2), Marta(12/2), Pierre(19/2), Marco(26/2)
7  Magnet Tests completed () ; 

4 Dipoles tests (2394 failed HV @CDBP, 2197 was only Cryo leak test)


1 Arc-SSS (  099 )

2 500-series ( from last year)

Situation from Cryogenics:  J-P Lamboy 
Situation  all SSS  : J Halik : 
No spares on 500 series so 523 ( & for remaining  500-series) problems  still imply “ use as” 
General Points:
· Feb working

· Eric, Kevin, GHH, Florence, JeromeA  absent 2 Feb to 18 Feb 2007
· A.O.B
Tests : s099 & 2394 done in 1st week. Other magnets done from last year(519,516)




2197 only cryo leak test;        so effectively only 2268 & 2407 done as real new magnet tests  

	Magnets under Test from 14-01-2007 to 20-01-2007 

	-
	Bench
	Magnet
	Shafts
	+
	Start Date
	Connect
Time
	Cooling
Time
	Cold Test
Time
	Warming
Time
	Disconnect
Time
	End Date
	Total
	Signed for
	
	

	1
	TBA2
	MBAR2268 
	[image: image7.png]



	


	21-12-2006 
	520 h.
	75 h.
	30 h.
	18 h.
	47 h.
	19-01-2007 
	690 h.
	Stripping
	
	

	2
	TBB1
	SSS519 
	[image: image9.png]



	


	08-12-2006 
	106 h.
	17 h.
	92 h.
	9 h.
	679 h.
	15-01-2007 
	903 h.
	Stripping
	
	

	3
	TBB2
	SSS516 
	[image: image11.png]



	


	15-12-2006 
	53 h.
	15 h.
	41 h.
	12 h.
	448 h.
	15-01-2007 
	569 h.
	Stripping
	
	

	4
	TBC1
	SSS099 
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	11-01-2007 
	20 h.
	18 h.
	12 h.
	62 h.
	50 h.
	18-01-2007 
	162 h.
	Stripping
	
	

	5
	TBE2
	MBBR2394 
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	16-12-2006 
	546 h.
	30 h.
	41 h.
	28 h.
	128 h.
	19-01-2007 
	773 h.
	Stand By
	
	

	6
	TBF1
	MBBL2197 
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	19-12-2006 
	520 h.
	23 h.
	9 h.
	20 h.
	164 h.
	19-01-2007 
	736 h.
	Stripping
	
	

	7
	TBF2
	MBAR2407 
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	19-12-2006 
	496 h.
	24 h.
	28 h.
	19 h.
	97 h.
	16-01-2007 
	664 h.
	Stripping
	
	


 

	-
	Magnet
	Q1
	Q2
	Q3
	Q4
	Q5
	Q6
	Q7
	Q8
	Q9
	Q10
	Q11
	Q12
	Q13

	1
	MBAR2268
	11916.5 A.
(D2_L)
	12507.6 A.
(D1_L)
	0 A.
(No Data)
	12849.7 A.
(D2_L)
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	-
	-
	-
	-
	-
	-
	-
	-

	2
	SSS519
	60 A.
(No Data)
	8905.3 A.
(QE_34)
	10002.9 A.
(QI_12)
	10351.9 A.
(QE_12)
	10485.5 A.
(QI_34)
	10527.7 A.
(QI_12)
	10579.8 A.
(QE_34)
	10619 A.
(QI-34)
	10682 A.
(QI_12)
	10710.3 A.
(QI_12)
	10732.1 A.
(QI_12)
	10826.1 A.
(QI_12)
	10879 A.
(QE_34)
	10908.2 A.
(QE_12)
	10921 A.
(QI_12)
	10913.3 A.
(QE_12)
	10968.9 A.
(QI_12)
	10989.2 A.
(QI_34)
	10968 A.
(QI_34)
	11000 A.
(No Data)
	11000 A.
(No Data)
	[image: image22.png]




	3
	SSS516
	11646 A.
(QI_34)
	11783 A.
(QE_12)
	12040.1 A.
(QE_12)
	12248 A.
(QE_34)
	12404 A.
(QI_12)
	12511 A.
(QE_12)
	12684 A.
(QI_12)
	12774 A.
(QE_34)
	12847.1 A.
(QI-12)
	12851 A.
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	-
	-
	-
	
	
	
	
	
	
	
	
	

	4
	SSS099
	12471 A.
(QI_34)
	12850 A.
[image: image24.jpg]



	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	

	5
	MBBR2394
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	

	6
	MBBL2197
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	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	

	7
	MBAR2407
	11657.8 A.
(D1_U)
	12258 A.
(D1_L)
	12641.3 A.
(D2_L)
	0 A.
(No Data)
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	-
	-
	-
	-
	-
	-
	-
	-
	
	
	
	
	
	
	
	
	


Back to Homepage | For print (Test/Quench/Graph)| For print (Waiting time\Disconnect time\Graph)

TERMINATION COUNTER - 2007
	No
	MAGNET
	BENCH
	ARRIVAL
	DEPARTURE
	Comments

	1
	2414
	E1
	12-12-2006
	
	

	2
	2408
	F2
	16-01-2007
	
	

	3
	2423
	E2
	19-01-2007
	
	

	4
	2328
	F1
	19-01-2007
	
	

	5
	2425
	
	
	
	

	6
	2424
	
	
	
	

	7
	2326 (MM)
	
	
	
	

	8
	2267 (MM)
	
	
	
	

	9
	2263
	
	
	
	

	10
	2426
	
	
	
	

	11
	2324
	
	
	
	NC on capillary

“might” come again 

	12
	2790
	
	
	
	Spare

	13
	2624
	
	
	
	Spare

	14
	2598
	
	
	
	Spare

	15
	2420
	
	
	
	Spare

	16
	2419
	
	
	
	Spare

	17
	2418
	
	
	
	Spare

	18
	2399(MM)
	
	
	
	Spare

	19
	2551
	
	
	
	Spare

	20
	2739
	
	
	
	Spare


12 of  20 Dipoles for completion of Tunnel                            Updated : January 31, 2007
Note : 

1)   The red line shows the completion of magnets for tunnel.

2)   MBBR 2394 is under investigation for HV problem at Cold (Richard) .  It will be replaced by a spare if this problem could’nt be solved. 

3)   One of  xxxx will replace 2324 ?
SICK DIPOLES

	 No
	MAGNET
	Problem due to

	1
	2394
	HV at Cold

	2
	1334
	CDBP: QH Vs Gnd

	
	
	


TERMINATION COUNTER - 2007
	23
	SSS 523
	D1
	10-01-2007
	
	?? correctors ??
MQ int-turn short ?

	24
	SSS 531
	
	
	
	19-01-2007*

	25
	SSS 529
	
	
	
	24-01-2007*

	26
	SSS 530
	
	
	
	01-02-2007*

	27
	SSS 361
	B1
	16-01-2007
	
	

	28
	SSS 356
	
	
	
	23-01-2007*

	29
	SSS 374
	
	
	
	26-01-2007*


 7 of  7 SSSs for completion of Tunnel                                    Updated : January 31, 2007
* Expected Arrival Date ( Bldg 904 information)

SICK SSS

	No
	MAGNET
	Problem due to

	1
	
	

	2
	
	

	
	
	


	Date
	Bench
	Problem
	Type
	Application/

soft
	Action/advise

	Wednesday_24_M
	TBB1
	SSS361: While launching the MM2 power supply communication error has occurred.
	MTM
	
	Guy informed, he will check it & also he will re- launch the test

	Wednesday_24_A
	TBF1
	MB2328: MB detrained during second training by 324 Amp. 
	MTM
	
	As per the de-training rules since value is still >12000 A so we can follow the normal training rule chart. Hence PT8, PT9 and then warm up quench.

	Thursday_25_M
	TBB1
	Could not launch MM. MMP Error message "You need a power supply!"

The PLCSE on all clusters not working. 
	MTM
	
	Guy informed about MM problems. He found that the water PLC went to standby mode. Solved.

	Thursday_25_M
	TBF1
	MB2328: OP team tried to reset TEMA on F1. Error message "...... UDF socket"

	AB/CO
	TEMA
	Giorgio informed about TEMA problem.Solved


	Thursday_25_M
	All
	Problem with e-logbook and e-traveller (Jurek could not sign it).


	
	
	Solved by J.M Elyn (e-logbook) and Giorgio’

	Friday_26_A
	TBA2
	MB2326: Powering of CD EF corrector in the dipole.

Inductance value missing.
	MTM
	
	Stéphane and Guy contacted. Giorgio will power the correctors Monday.

	Friday_26_A
	TBE1
	MB2424: During  LF heater discharge (IAP at warm), VT_anode, VT_cathode  errors (-1 V). 
	MTM
	
	Salem found that :

· the connector were connected wrongly

· the dessicant bag was present in the IFS box.

	Friday_26_A
	TBF2
	MB2267: Magnet connected by MTM team. No WP04 done.
	MTM
	
	Stéphane asked to go on and check fater the connection the electrical integrity.

	
	
	
	
	
	

	
	
	
	
	
	

































































